# Vertical integration of general practices with acute hospitals in England: rapid impact evaluation

## Manbinder Sidhu,<sup>1</sup> Catherine L Saunders,<sup>2</sup> Charlotte Davies,<sup>3</sup> Gemma McKenna,<sup>1</sup> Frances Wu,<sup>3</sup> Ian Litchfield,<sup>4</sup> Fifi Olumogba<sup>3</sup> and Jon Sussex<sup>3\*</sup>

<sup>1</sup>University of Birmingham, Health Services Management Centre, Birmingham, UK <sup>2</sup>Primary Care Unit, Department of Public Health and Primary Care, University of Cambridge, Cambridge, UK <sup>3</sup>RAND Europe, Cambridge, UK <sup>4</sup>University of Birmingham, Institute of Applied Health Research, Birmingham, UK

\*Corresponding author jsussex@randeurope.org

## **Disclosure of interests of authors**

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/PRWQ4012.

Primary conflicts of interest: None.

Published October 2023 DOI: 10.3310/PRWQ4012

# Plain language summary

Vertical integration of general practices with acute hospitals in England: rapid impact evaluation

Health and Social Care Delivery Research 2023; Vol. 11: No. 17 DOI: 10.3310/PRWQ4012

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain language summary**

Our previous evaluation of when National Health Service hospitals take over the running general (medical) practices, based mainly on interviews with National Health Service staff, found that a major aim was to keep open general practices that might otherwise close. The follow-up study reported here looked at wider questions and drew on a wider range of evidence. Does a hospital-run practice affect patients' satisfaction with their care and how often they go to hospital? Does the arrangement work differently for patients with multiple long-term conditions? To answer these questions, we compiled the first ever list of all the general practices in England run by National Health Service hospitals: they total a little over 1% of all general practices. We found that general practices run by hospitals are not typical. We should not assume that the results found there would be repeated elsewhere if there were to be a widespread move to having hospitals run general practices.

Our research drew on 7 years of national data on patients' use of hospitals and 10 years of findings from the national General Practitioner Patient Survey. We also interviewed patients and staff at three locations in England where hospitals run general practices.

Patients' experience of, and satisfaction with, the care they receive from a general practice is similar whether it is run by a hospital or not. There are modest and temporary (1-2%) reductions in patients' accident and emergency department and outpatient attendances when a hospital starts running their general practice and larger and more permanent reductions (3-5%) in rates of emergency admissions and readmissions to hospital. Patients with multiple long-term conditions are affected similarly to other patients. The most important condition for success seems to be building and maintaining trust between clinicians and other staff at the hospital and the general practices.

## **Health and Social Care Delivery Research**

ISSN 2755-0060 (Print)

ISSN 2755-0079 (Online)

Health and Social Care Delivery Research (HSDR) was launched in 2013 and is indexed by Europe PMC, DOAJ, INAHTA, Ulrichsweb<sup>™</sup> (ProQuest LLC, Ann Arbor, MI, USA), NCBI Bookshelf and MEDLINE.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

This journal was previously published as Health Services and Delivery Research (Volumes 1–9); ISSN 2050-4349 (print), ISSN 2050-4357 (online)

The full HSDR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr.

#### Criteria for inclusion in the Health and Social Care Delivery Research journal

Reports are published in *Health and Social Care Delivery Research* (HSDR) if (1) they have resulted from work for the HSDR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HSDR** programme

The HSDR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HSDR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-and-social-care-delivery-research.htm.

#### This report

The research reported here is the product of an HSDR Rapid Service Evaluation Team, contracted to undertake real time evaluations of innovations and development in health and care services, which will generate evidence of national relevance. Other evaluations by the HSDR Rapid Service Evaluation Teams are available in the HSDR journal.

The research reported in this issue of the journal was funded by the HSDR programme or one of its preceding programmes as project number NIHR135618. The contractual start date was in May 2022. The final report began editorial review in March 2023 and was accepted for publication in June 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HSDR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HSDR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, these of the authors, those of the NHS, the NIHR, the HSDR programme or the Department of Health and Social Care.

Copyright © 2023 Sidhu *et al.* This work was produced by Sidhu *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

#### **BRACE: The Birmingham, RAND and Cambridge Rapid Evaluation Centre**

The BRACE Rapid Evaluation Centre (National Institute for Health and Care Research Birmingham, RAND and Cambridge Evaluation Centre) is a collaboration between the Health Services Management Centre at the University of Birmingham, the independent research organisation RAND Europe, the Department of Public Health and Primary Care at the University of Cambridge, and National Voices. BRACE carries out rapid evaluations of innovations in the organisation and delivery of health and care services. Its work is guided by three overarching principles:

- 1. Responsiveness. Ready to scope, design, undertake and disseminate evaluation research in a manner that is timely and appropriately rapid, pushing at the boundaries of typical research timescales and approaches, and enabling innovation in evaluative practice.
- 2. Relevance. Working closely with patients, managers, clinicians and health care professionals, and others from health and care, in the identification, prioritisation, design, delivery and dissemination of evaluation research in a co-produced and iterative manner.
- 3. Rigour. All evaluation undertaken by the team is theoretically and methodologically sound, producing highly credible and timely evidence to support planning, action and practice.

## NIHR Journals Library Editor-in-Chief

#### Dr Cat Chatfield Director of Health Services Research UK

### **NIHR Journals Library Editors**

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

**Dr Peter Davidson** Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Dr Rob Riemsma** Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk